Author's response to reviews

Title: Children eat their school lunch too quickly: an exploratory study of the effect on food intake

Authors:

Modjtaba Zandian (modjtaba.zandian@ki.se)
Ioannis Ioakimidis (ioannis.ioakimidis@ki.se)
Jakob Bergstrom (jakob.bergstrom@ki.se)
Ulf Brodin (ulf.brodin@ki.se)
Cecilia Bergh (cecilia.bergh@mando.se)
Julian Shield (j.h.p.shield@bristol.ac.uk)
Per Sodersten (per.sodersten@ki.se)

Version: 2 Date: 6 September 2011

Author's response to reviews:

Dear Editor

Thanks for the advise to expand on the Background section, which I have done. I have re-listed the references and added a reference (26) to the recent paper in the Lancet pointing out the importance of environmental factors for food intake particularly amongst children.

I hope that the manuscript can now be sent for peer review.

Yours sincerely,

Per Sodersten
Karolinska Institutet